Author's response to reviews

Title: Psychometric behaviour of the Strengths and Difficulties Questionnaire (SDQ) in the Spanish National Health Survey 2006

Authors:

Manuel Gómez-Beneito (Manuel.Gomez-Beneyto@uv.es)
Andreu Nolasco (nolasco@ua.es)
Joaquín Moncho (joaquin.moncho@ua.es)
Pamela Pereyra-Zamora (pamela.pereyra@ua.es)
Nayara Tamayo-Fonseca (nayara.tamayo@ua.es)
Manuel Girón (giron@icali.es)
Mikel Munárriz (mikelmf@comcas.es)
José Salazar (jose.salazar@eresmas.net)
Rafael Tabarés-Seisdedos (rafael.tabares@uv.es)

Version: 2 Date: 10 May 2012

Author's response to reviews:

In response to your comments, we inform you the following:

Ethics: We think that we do not need specific ethical approval because we used secondary data derived from a survey carried out by the Spanish Ministry of Health, Social Politics and Equality and the Spanish National Statistical Institute, which are the highest authorities on health and statistics in Spain. Ethical considerations were taken into account by these authorities.

Consent: In the methods section, we have included a sentence and a reference where the reader can clarify the details of the methodology of the original survey (sample design, sample size, sampling procedure, consent,...)

We hope we have answered your comments

Best wishes
Andreu Nolasco